Author's response to reviews

Title: A survey of attitudes and factors associated with successful CPR knowledge transfer in an older population most likely to witness cardiac arrest: design and methodology

Authors:

Christian Vaillancourt (cvaillancourt@ohri.ca)
Jeremy Grimshaw (jgrimshaw@ohri.ca)
Jamie C Brehaut (jbrehaut@ohri.ca)
Martin Osmond (Osmond@cheo.on.ca)
Manya L Charette (mancharette@ohri.ca)
George A Wells (gawells@ottawaheart.ca)
Ian G Stiell (istiell@ohri.ca)

Version: 3 Date: 20 October 2008

Author's response to reviews:

All major and minor revisions were made in the abstract and manuscript as requested.